Dear editor of BMC Family Practice,

Thank you for the option to resubmit our manuscript entitled: ‘Development and evaluation of a blended educational programme for general practitioners’. We believe that the reviewers’ comments helped us to improve our manuscript. Please find our response below this letter. We think this paper is of particular interest to your readers, and we hope you reconsider our manuscript for publication in BMC Family Practice.

With kind regards,

On behalf of all authors, Ivo K. Joore, MSc, PhD

Department of General Practice/ Family medicine

Academic Medical Center - University of Amsterdam
Response to the reviewers

Question 1. Please remove the response to reviewers from the file inventory, as it is no longer needed at this stage of the editorial process.

Answer: We removed this file.

Question 2. Please list email addresses for all authors on the title page.

Answer: We included the email addresses.

Question 3. Please move the keywords section after the Abstract.

Answer: We moved the keywords section after the abstract.

Question 4. Please revise the "Introduction" heading to "Background" to meet journal formatting guidelines.

Answer: We changed Introduction into background.

Question 5. Please include a list of abbreviations used in the manuscript before the Declarations section.

Answer: We included this in the manuscript.
List of abbreviations

HIV: human immunodeficiency virus; GP: general practitioner; AIDS: acquired immunodeficiency syndrome; STI: sexually transmitted infection; AMC-UvA: Academic Medical Center, University of Amsterdam; MSM: Men who have sex with men; UK: United Kingdom; IRR: incidence rate ratio; CI: Confidence interval; H-TEAM: HIV Transmission Elimination Amsterdam

Question 6. Please include a "Declarations" heading before the Declarations section.

Answer: We included this in the manuscript.

Question 7. Please include the sentence "All authors have read and approved the final version of this manuscript" in the Author contributions section and ensure that this is the case.

Answer: We included this in the manuscript.

Question 8. Please note that author A van der Maat is not included in the Author contribution section. Please specify the individual contributions of all authors in the manuscript.

Answer: We included the author A van der Maat in the Author contribution section and specified even more the individual contributions of all authors in the manuscript.

Question 9. In the Funding section, please indicate the role of the funding body in the design of the study and collection, analysis, and interpretation of data and in writing the manuscript.

Answer: We included the sentence “Aids Fonds had no role in the design of the study and collection, analysis, and interpretation of data, or in writing the manuscript.”

Question 10. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables and figures should also be clean versions.

Answer: We only included the relevant documents.